Author’s response to reviews

Title: Estimating the annual risk of infection with Mycobacterium tuberculosis among adolescents in Western Kenya in preparation for TB vaccine trials

Authors:

Videlis Nduba (vnduba@gmail.com)

Anna van’t Hoog (a.h.vanthoog@aighd.org)

Annefleur de Bruijn (annefleurdebruijn@hotmail.com)

Ellen Mitchell (mitchell@iss.nl)

Kayla Laserson (kayla.laserson@gatesfoundation.org)

Martien Borgdorff (m.w.borgdorff@amc.uva.nl)

Version: 4 Date: 20 Jul 2019

Author’s response to reviews:

Editors comments:

1. please add a conclusions section: This has been added

2. Please include the full name of the ethics committee, institute and reference number: This has been added in section on declarations

3. Please confirm informed parental consent, written or verbal was obtained from all parents/guardians- This has been confirmed that written informed parental consent was obtained from all parents/guardians.

4. In the funding section of your declarations clarify role of the funding body in the design of the study- This has been clarified. The funding bodies had no role in design, analysis and writing of the manuscript.

5. Please amend statement Role of Funder to Funding: this has been amended.

6. Please add a section " additional files" This has been added